Dear Sir/Madam

This is an interesting case report of a middle age postman presenting with atypical chest pain and lipaemia retinalis and found to have severe hypertriglyceridaemia. This case report would be interesting to the readers because it highlights the complexity of hyperlipidaemia and the different presentation of such uncommon form of familial type of hyperlipidaemia. Besides this case report support the role of combined lipid modifying agents in lowering triglycerides and cholesterol as well as the paradoxical improvement in liver enzymes.

Dr Ali Al-bahrani MBCh.B, MSc, MRCPath
Dept Clinical Biochemistry
Royal Liverpool University Hospital.
Duncan Building, 4th floor
L69 3GA